## SPIRE-RAL-MOM-001173

## Notes on project managers teleconference 11/02/02

Chair: Eric Sawyer

**Present:** 

Lionel Duband	X
Ben Parvin	
Berend Winter	
Chris Brockley-Blatt	X
Matt Griffin	
Peter Hargrave	X
Jerry Lilienthal	
Joe Taylor	X
Ian Pain	X
Don Peterson	X
Bruce Swinyard	
Ken King	
Dominique Pouliquen	
Gary Parks	
Jean-Louis Auguerres	
Doug Griffin	
Dave Smith	
Eric Clark	
John Delderfield	
Didier Ferrrand	
Riccardo Cerulli-Irelli	
Renato Orfei	
Brian Stobie	
Fredericke Pinsard	

## General

Note on subsystem requirements for IBDR presentations will be issued soon.

## **Subsystem reports**

IFSI (Riccardo) – DPU Not present

CEA Grenoble (Lionel) – Cooler Large test cryostst delivered and will be commissioned. Cooler centre parts now 80% delivered, EB welding this week Heat switches due this week. Kevlar test cord has broken after about 8 million. A higher tension (20% increase) test will now be conducted.

USK. (Don) - Shutter Harness definition doc being reviewed. Funding meeting expected this week.

Cardiff. (Peter) Filters, calibrators

ICDs issued.

300mK filters are almost ready, thermal cycle test complete, ship to JPL this week. Spectrometer Lens design being progressed

Mk 2 300mK strap support has been delayed but should be delivered this week...

ATC (Ian) - BSM ICD update. Cold test delayed. Representative wire source required, Doug helping.

JPL (Jerry) - Detectors Not present

MSSL (Chris) – Structure
Main optical bench out for manufacture.
2K box (photometer) being machined.
Analysis of A frames continuing.
Discussion on lubrication of screws required.
Harness details of SMEC required ASAP.
Backshell details of spectrometer BDA required ASAP.
Thermal busbar test rig in manufacture.

LAM. (Dominique) - Mirrors, SMEC Not present

SAp (Jean-Louis) - DRCU Not present

RAL, (Dave) - AIT Not present

No.	Action	Who	Due date	Response/Comment	Status
•					
·					